Author's response to reviews

Title: Perceptions of treatment for tics among young people with Tourette syndrome and their parents: A mixed methods study

Authors:

José Cuenca (jose.cuenca@nottingham.ac.uk)
Cris Glazebrook (cris.glazebrook@nottingham.ac.uk)
Tim Kendall (tim2.kendall@virgin.net)
Penny Bunton (penny.bunton@manchester.ac.uk)
Tammy Hedderley (tammy.hedderly@gstt.nhs.uk)
Isobel Hayman (i.heymann@ucl.ac.uk)
Georgina Jackson (georgina.jackson@nottingham.ac.uk)
Tara Murphy (tara.murphy@gosh.nhs.uk)
Hugh Rickards (hugh.rickards@bsmhft.nhs.uk)
Mary Robertson (rejummr@ucl.ac.uk)
Jeremy Stern (jstern@tourettes-action.org.uk)
Chris Hollis (chris.hollis@nottingham.ac.uk)

Version: 2
Date: 26 January 2015

Author's response to reviews:

A cover letter has been uploaded describing revisions to the manuscript